Author’s response to reviews

Title: Glycated Hemoglobin Versus Oral Glucose Tolerance Test in The Identification of Subjects With Prediabetes in Qatari Population

Authors:

Muhammad Abdul-Ghani (abdulghani@uthscsa.edu)
Saadallah Iskandar (IAKANDAR@QP.COM.QA)
Ayman Megahid (YMNDODO20@GMAIL.COM)
Dalia Kamal (DKAMAL@HAMAD.QA)
Osama Migahid (PREDRMIGAHID@GMAIL.COM)
Ralph DeFronzo (ALBARADOD@UTHSCSA.EDU)
Amin Jayyousi (AJAYYOUSI@HAMAD.QA)
Mahmood Al Jaidah (ALJAIDAH@QP.COM.QA)

Version: 4 Date: 11 Jul 2019

Author’s response to reviews:

1- The 'Consent for publication' section covers manuscripts that contain any individual person’s data in any form (including individual details, images or videos), for which consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian.

If your manuscript does not contain any individual person’s data, please state “Not applicable” in this section.

The consent for publication has been revised to:” Not applicable”

2- Please note if you wish to acknowledge someone by their full name in the Acknowledgements, please ensure you have obtained permission from them to so do.

We obtained the consent of people acknowledged in the manuscript
3- Figures 1 & 2 are missing from the file inventory; please upload them.

Figures 1 and 2 have been added

4- At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the information in the designated input box only.

A clean version of the manuscript has been added as well as figures 1 and 2